Author's response to reviews

Title: The oral health of heroin drug users: case study in Bosnia and Herzegovina

Authors:

Zorica J Terzic Supic (zorica.terzic-supic@mfub.bg.ac.rs)
Ranko Petrovic (rankopetrovic@yahoo.com)
Milena Santric Milicevic (msantric@med.bg.ac.rs)
Goran Trajkovic (t_goran@med.bg.ac.rs)
Zoran Bukumiric (zoran.bukumiric@mfub.bg.ac.rs)

Version: 2 Date: 19 July 2013

Author's response to reviews:

Please find enclosed the paper "The oral health of heroin drug users: case study in Bosnia and Herzegovina". Bosnia and Herzegovina is the country on the Balkan Drug Route which, in the post-war period, faced the problem of increased drug use, with insufficient capacity for fight against it. This was the first time that the research of biological and behaviours among injection drug users was run. The survey included their oral health too. The research on oral health among heroin drug users has never been run before, not only in Bosnia and Herzegovina but in whole region and other countries in Europe. Generally, this kind of research is rarely led anywhere. The cross sectional survey was a part of the UNICEF's research Biological and behavioural survey among injection drug users in Sarajevo, Banja Luka and Zenica in Bosnia and Herzegovina.